Author's response to reviews

Title: Association study between vitamin D receptor gene polymorphisms and asthma in the chinese han population: a case-control study

Authors:

   Ahlem Saadi (ahlem_sa@hotmail.com)
   Guimin Gao (gaoguimin@sdu.edu.cn)
   huaichen Li (lihuaichen@163.com)
   Chunhua Wei (weichunhua@163.com)
   Yaoqin Gong (yxg8@sdu.edu.cn)
   Qiji Liu (liuqiji@sdu.edu.cn)

Version: 4 Date: 15 July 2009

Author's response to reviews: see over
Dear editor,

Thanks for accepting our manuscript. According to the suggested revision, we have made a careful check and changes including table format and figure. Also, each word was checked to make sure no typographical errors. We have deleted reference 26 for we have forgot deleted it in the last revising. Many thanks for your hard work for processing the paper.

Best regards

Qiji Liu
Institute of Medical Genetics,
School of Medicine, Shandong University
Jinan 250012
P.R.China
Tel:86-531-88382043
Fax:86-531-88382115
E-mail: liuqiji@sdu.edu.cn